Author's response to reviews

Title: Factor VII deficiency and developmental abnormalities in a patient with partial monosomy of 13q and trisomy of 16p: case report and review of the literature

Authors:

Brian P Brooks (brooksb@mail.nih.gov)
Jeanne M Meck (meckj@gunet.georgetown.edu)
Bassem R Haddad (haddadb1@georgetown.edu)
Claude Bendavid (claude.bendavid@univ-rennes1.fr)
Delphine Blain (dblain@mail.nih.gov)
Jeffrey A Toretsky (jat42@georgetown.edu)

Version: 6 Date: 19 December 2005

Author's response to reviews:

Dear Editors:

Thank you for accepting our manuscript, "Factor VII deficiency and developmental abnormalities in a patient with partial monosomy of 13q and trisomy of 16p: case report and review of the literature." As requested by the editors, we have made the following changes:

1. We have changed the title in the online submission to match that of the written manuscript.
2. We have changed the authors' names in the online submission to match that of the written manuscript.
3. We have deleted the header
4. We have changed the capitalization of the title
5. We have removed the running title from the title page
6. We have removed the keywords from the title page.
7. We have eliminated the underlining
8. We have changed the title to "Authors' contributions."